Author's response to reviews

Title: Grp78 promotes the invasion of hepatocellular carcinoma

Authors:

Rongjian Su (rongjiansu@hotmail.com)
Zhen Li (lizh9@163.com)
Hongdan Li (lihongdan101@hotmail.com)
Huijuan Song (shi-shi@tom.com)
Cuifen Bao (baocuifen@yahoo.com.cn)
Jia Wei (volve@gmail.com)
Liufang Cheng (liufang_cheng@yahoo.com.cn)

Version: 3 Date: 15 January 2010

Author's response to reviews: see over
Dear editor:

Sorry to bother you.

We have revised our manuscript as your requirements.

1. The pictures have been re-processed. All the pictures under the same index have been merged as one picture.

2. The authors’ name has been re-written as your requirement. For example Su Rong-jian has been revised as Rongjian Su.

3. We have declared that all the authors read and approved the final manuscript in the section of author’s contribution.

4. We have re-processed the tables; all the merged cells have been displayed.

5. As the editor’s requirement, we have revised $p$ value in table 3 as $<0.01$.

6. We have checked the final manuscript and corrected all the typing errors as we can.

Best wishes

Sincerely

Rongjian Su